Author's response to reviews

Title: Nutrition promotion approaches preferred by socioeconomically disadvantaged Australian adolescents: a qualitative study

Authors:

Lena D Stephens (l.stephens@deakin.edu.au)
Sarah A McNaughton (sarah.mcnaughton@deakin.edu.au)
David Crawford (david.crawford@deakin.edu.au)
Kylie Ball (kylie.ball@deakin.edu.au)

Version: 2 Date: 17 September 2014

Author's response to reviews: see over
Dear Editor,

Please find attached a copy of our manuscript entitled “Nutrition promotion approaches preferred by socioeconomically disadvantaged Australian adolescents: a qualitative study”. We would be grateful if you would consider this manuscript for publication in *BMC Pediatrics*.

This paper reports on a qualitative study exploring socioeconomically disadvantaged adolescents’ perceptions regarding potential strategies to improve eating behaviours and preferred nutrition promotion dissemination methods. Findings provide rich insight into strategies and avenues for distributing nutrition promotion messages and strategies to improve eating behaviours among this target group at high risk of poor diets. Given the increasing rates of socioeconomic disparities in adolescent health worldwide, we are confident the manuscript would be of strong interest to the Journal’s readership.

Please note the authors declare that no conflicts of interest exist.

The following individuals are proposed as potential reviewers of this manuscript:

1. **Name:** Associate Professor Elaine Molaison  
   **Address:** Department of Nutrition and Food Systems, Fritzscche-Gibbs (FG), 210, The University of Southern Mississippi, 118 College Drive, Box #5172, Hattiesburg, MS 39406, USA.  
   **Email:** Elaine.Molaison@usm.edu

2. **Name:** Professor Diane Neumark-Sztainer  
   **Address:** Division of Epidemiology and Community Health, University of Minnesota, 1300 S. 2nd Street, Suite 300, Minneapolis, MN 55454, USA.  
   **Email:** neumark@epi.umn.edu

3. **Name:** Associate Professor Jayne Fulkerson  
   **Address:** School of Nursing, University of Minnesota, 5-140 Weaver-Densford Hall, 308 Harvard Street SE, Minneapolis, MN 55455, USA.  
   **Email:** fulke001@umn.edu

We look forward to hearing from you about the outcome of this submission.

Yours sincerely,

Lena Stephens, PhD

Mail: Centre for Physical Activity and Nutrition Research, School of Exercise and Nutrition Sciences, Deakin University, 221 Burwood Highway, Burwood VIC 3125, AUSTRALIA

Email: l.stephens@deakin.edu.au

Phone: +61-3-9244-6328

Fax: +61-3-9244-6017